Author’s response to reviews

Title: Continuity of care is an important and distinct aspect of childbirth experience: findings of a survey evaluating experienced continuity, experienced quality of care and women's perception of labor.

Authors:

Hilde Perdok (Hildeperdok@gmail.com)
Corine Verhoeven (c.verhoeven@vumc.nl)
Jeroen van Dillen (jeroen.vandillen1@radboudumc.nl)
Tjerk Jan Schuitmaker (t.j.schuitmaker@vu.nl)
Karla Hoogendoorn (kahoogendoorn@hotmail.com)
Jolanda Colli (j.colli@verloskundigenleo.nl)
François Schellevis (f.schellevis@nivel.nl)
Ank de Jonge (ank.dejonge@vumc.nl)

Version: 2 Date: 27 Sep 2017

Author’s response to reviews:

Dear Editor,

We have read the comments of the editor and we have revised the title of the manuscript accordingly.

'Continuity of care is an important and distinct aspect of childbirth experience: findings of a survey evaluating experienced continuity of care, experienced quality of care and women's perception of labor.'

We hope the manuscript is suitable for publication in your journal.

On behalf of the authors,

Sincerely,

Hilde Perdok